Author's response to reviews

Title: Optic atrophy, necrotizing anterior scleritis and keratitis presenting in association with Streptococcal Toxic Shock Syndrome: a case report

Authors:

Konstantinos I Papageorgiou (papageorgiouk@doctors.org.uk)
Alexander S Ioannidis (alexioannidis@hotmail.com)
Petros S Andreou (pandreou@aol.com)
Ajay Sinha (Ajay.Sinha@meht.nhs.uk)

Version: 6 Date: 29 February 2008

Author's response to reviews:

Dear Editor,

the requested changes have been updated:
The authors qualifications have been removed and their emails have been added
The format of the abstract has been changed as requested
The references have been changed in order to include all the names of the contributing authors
The consent related paragraph has been changed to the requested format
The order of apperance in the manuscript of the abbreviations, consent and competing interests sections has been updated

Kind Regards

Dr KI Papageorgiou